Author's response to reviews


Authors:

Catherine Pope (cjp@soton.ac.uk)
Susan Halford (Susan.Halford@soton.ac.uk)
Joanne Turnbull (j.c.turnbull@soton.ac.uk)
Jane Prichard (J.S.Prichard@soton.ac.uk)
Melania Calestani (M.Calestani@soton.ac.uk)
Carl May (crm@soton.ac.uk)

Version: 2 Date: 15 February 2013

Author's response to reviews:

Dear Sir/Madam

I am pleased to submit a revised version of our paper and would like to express our thanks to the referees and editors for their careful reading of the paper and the very helpful comments and suggestions made. Please see attached document for details of the revisions.

Catherine Pope